Author’s response to reviews

Title: Prevalence of Gingivitis and Calculus in 12-year-old Puerto Ricans: A Cross-Sectional Study

Authors:

Augusto Elias-Boneta (augusto.elias@upr.edu)
Karol Ramirez (karol.ramirez@ucr.ac.cr)
Sona Rivas-Tumanyan (sona.tumanyan@upr.edu)
Margarita Murillo (margarita.murillo@upr.edu)
Milagros Toro (milagros.toro@upr.edu)

Version: 5 Date: 15 Dec 2017

Author’s response to reviews:

Response to the Editor

1. Author Change Forms

Thank you for submitting your revised manuscript, the revisions all look good. Unfortunately we are unable to accept the submitted author change form as the signatures have been digitally added. Please fill in the attached form and have every author sign it, both those on the paper and those added/removed. If authors are located in other cities/countries/etc. it is acceptable for an author to print off their own form, sign it and have multiple sheets compiled before returning, but please be sure every author has signed for themselves. Please not you manuscript cannot be accepted until this form has been completed.

Response:

Since some of the authors are located in other cities/countries, we are submitting the change of authorship request form including 4 Declarations of agreement (Section 6) pages. All the authors have signed by themselves.

2. List of Abbreviations

Please move this section so that it precedes the Declaration

Response:
The List of Abbreviations have been moved before the Declaration.